Author’s response to reviews

Title: Relationships between cognitive function and body composition among community-dwelling elderly: a cross-sectional study

Authors:

Hye-Mi Noh (noham111@hanmail.net)
Sohee Oh (oh.sohee@gmail.com)
Hong Ji Song (hongjisong5@gmail.com)
Eun Young Lee (euny9930@hanmail.net)
Jin-Young Jeong (jjy@hallym.ac.kr)
Ohk-Hyun Ryu (ohryu30@hallym.or.kr)
Kyung-Soon Hong (kshong@hallym.or.kr)
Dong-Hyun Kim (dhkims@hallym.ac.kr)

Version: 1 Date: 24 Aug 2017

Author’s response to reviews:

Dear Editor,

We greatly appreciate the reviewers’ specific suggestions and constructive comments on our Manuscript ID BGTC-D-16-00496, entitled "Relationships between cognitive function and body composition among community-dwelling elderly: a cross-sectional study". We redesigned our study and reanalyzed the data according to your recommendation. Therefore it took long time to revise our manuscript. Details of how we have responded to each of the reviewer’s comments are provided in below response letter. We highlighted all the changes made in manuscript (tracked). We appreciate your consideration and look forward to hearing good news from you soon.

Yours sincerely,

Hong Ji Song, M.D., Ph.D.